Author's response to reviews

Title: Mental disorders in Bangladesh: A Systematic Review

Authors:

Mohammad D Hossain (ddrhossain@yahoo.com)
Helal U Ahmed (soton73@gmail.com)
Waziul A Chowdhury (profwaziul@squarehospital.com)
Louis W Niessen (lniessen@jhsph.edu)
Dewan S Alam (dsalam@icddrb.org)

Version: 2
Date: 21 November 2013

Author's response to reviews: see over
November 21, 2013

The Editor,
BMC Psychiatry
United Kingdom

Subject: Revised Manuscript for consideration for publication in the “BMC Psychiatry”

Dear Sir,

Enclosed please find attached herewith the revised manuscript entitled ‘Mental disorders in Bangladesh: A Systematic Review’ which has been updated with the missing information as suggested. Changes include the following:

Competing interests declaration on competing interest has been added right after the conclusion.

Author’s contributions have been included before the acknowledgments and references. Acknowledgments section has been included between the author’s contributions section and the reference list.

We are agreeable to make any further change in the text and/or tables if you feel it is necessary or suggest. We hope you would find this manuscript interesting to publish in your prestigious journal.

Thank you.

Yours sincerely,

Dr. Dewan S. Alam, MBBS, MMedSc, PhD.
Acting Director, Centre for Control of Chronic Diseases (CCCD)
&
Coordinator, Chronic Disease Epidemiology and Genetics RG

68, Shahid Tajuddin Ahmed Sharani
Mohakhali, Dhaka-1212, Bangladesh
Tel: +880 2 8818583 (direct)
+880 2 8860523-32 Ext'n 2521
Cell: +8801713093870
Fax: +880 2 8823963